

Response to Reviewer 1 Comments

Comment:

Reviewer #1:

Scientific Quality: Grade B (Very good)

Language Quality: Grade B (Minor language polishing)

Conclusion: Accept (General priority)

Specific Comments to Authors:

Thanks to the authors for submitting the Editorial on "Importance of the Gut Microbiota in the Gut-Liver Axis in Normal and Liver Disease." This topic of gut microbiota has garnered significant international interest among scholars in recent years. The Editorial offers an enlightening perspective on the gut microbiota from the viewpoints of healthy individuals and those with liver diseases, as well as discusses the bidirectional interactions between the gut microbiota and the liver. It is an insightful piece that merits publication and is highly recommendable.

Response: Thank you very much for your review of the manuscript and your opinion on its quality and content